Author's response to reviews

Title: Cognitive and psychosocial development of HIV pediatric patients receiving highly active anti-retroviral therapy: a case-control study

Authors:

Loretta Thomaidis (dr_thomaidis@yahoo.gr)
Georgia Bertou (drgbertou79@yahoo.com)
Elena Critselis (ecritselis@yahoo.com)
Vassiliki Spoulou (ppk@otenet.gr)
Dimitritios A Kafetzis (kafetzis@ath.forthnet.gr)
Maria Theodoridou (alexandratheo@yahoo.gr)

Version: 5 Date: 20 December 2010

Author's response to reviews:

Thank you for your comments.
We added 'between groups' to the end of the 10th line of the 'Methods'section of the Abstract in the submission system, so that it matches exactly the abstract in the manuscript.
We also removed the word count from the title page(p3)of the manuscript as suggested.
Thank you